Author’s response to reviews

Title: Clinical Predictors of Protracted Length of Stay in Ontario Complex Continuing Care Hospitals

Authors:

Luke Turcotte (luke.turcotte@uwaterloo.ca)

Chris Perlman (chris.perlman@uwaterloo.ca)

Brant Fries (bfries@umich.edu)

John Hirdes (hirdes@uwaterloo.ca)

Version: 2 Date: 26 Feb 2019

Author’s response to reviews:

1. Please provide a List of abbreviations after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

We have included a list of abbreviations.

2. Please provide a Keywords section after your Abstract that includes three to ten keywords representing the main content of the article.

We have included a keyword list.